Title: Leg elevation decreases the incidence of post-spinal hypotension in cesarean section: a randomized controlled trial

Authors:

Ahmed Hasanin (ahmedmohamedhasanin@gmail.com)
Ahmed Aiyad (dr.ahmed_aiyad@hotmail.com)
Ahmed Elsakka (ahmedsakka@hotmail.com)
Atef Kamel (Menna_atef01@yahoo.com)
Reham Fouad (rehamfouad@rocketmail.com)
Mohamed Osman (mosman55@yahoo.com)
Aly Mokhtar (alyaly4444@gmail.com)
Sherin Refaat (sherin.refaat@hotmail.com)
Yasmin Hassabelnaby (yalnaby@yahoo.com; yalnaby@kasralainy.edu.eg)

Version: 3 Date: 25 Mar 2017

Author’s response to reviews:

1- the form is completed and included as additional files and sent as separate email to editor
2- date is added
3- cover letter is removed
4- additional files include: registration, change authorship form
5- done